review of income and wealth

Review of Income and Wealth Series 64, Number 2, June 2018 DOI: 10.1111/roiw.12369

FOUR DECADES OF THE ECONOMICS OF HAPPINESS: WHERE NEXT?

BY ANDREW E. CLARK*

Paris School of Economics - CNRS

There has been explosive growth in the analysis of subjective well-being in Economics over the past 40 years. This article reviews some of this growth, and suggests a number of domains in which future research may proceed.

JEL Codes: D60, I31

Keywords: economics, subjective well-being

1. INTRODUCTION

Much has changed in Economics over the past 40 years. One such notable change that I would like to discuss here has been the remarkable rise in the interest shown by economists in subjective variables in general and in particular measures of subjective well-being.

While measures of subjective health have long been used in Economics, seemingly without any particular adverse comment, and subjective evaluations of job, life or income appeared as mainstays in Psychology and Sociology, Economics in the 1970s appeared to be virgin territory in this respect. There were a few notable exceptions to this general aversion to subjective well-being data, and these have actually become very well-cited (Easterlin, 1974; Hamermesh, 1977 and Freeman, 1978). But these contributions were rather treated as intellectual curios, and attracted only little attention from Economists in the 1970s and 1980s. In fact, the whole well-being literature seemed to be largely moribund in Economics for roughly 15 years from the end of the 1970s.

The economics of happiness really seemed to take off in the 1990s¹, and in a remarkable way. It is in particular worth underlining that as I write four of the 20 most-cited articles ever published in the *Economic Journal* explicitly have the

Note: This paper draws on my work with very many co-authors, to whom I wish to express my gratitude. An anonymous referee provided helpful comments, and Ed Diener graciously pointed me in the right direction for information on citations to subjective well-being articles. I thank Joyup Ahn, Danny Blanchflower and Conchita D'Ambrosio for useful advice, and seminar participants at the Korea Labor Institute's International Conference of Work and Happiness (Seoul).

 ${}^{1}As$ it did in other disciplines too. Diener *et al.* (2016) note that there were over 14,000 publications in 2015 on subjective well-being (SWB).

© 2018 The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

^{*}Correspondence to: Andrew Clark, PSE, 48 Boulevard Jourdan, 75014 Paris, France (Andrew. Clark@ens.fr).

word "happiness" in their title² (and two of the three most-cited articles in *Journal* of *Public Economics* deal with the question of subjective well-being).³

Happiness data have been used for broadly three different purposes, which I shall attempt to review in this article. At the first and simplest level, they were used as left-hand side variables in empirical analyses to answer the question: What makes people happy? The role of labor-force status and income were of particular interest among economists here. Second, the roles were reversed, with subjective well-being now appearing on the right-hand side of equations that were estimated to show what happy people did. The outcomes here ranged from income and productivity to labor-market behavior, marriage, children, morbidity and mortality (Freeman's 1978 paper showing that job satisfaction predicted individuals' future job quits in panel data was very much a precursor in this field).

Last, happiness data have been used to help address a number of economic puzzles, such as:

- If the self-employed are happier than the employed, why do most of us still work for other people? (Blanchflower and Oswald, 1998 and Frey and Benz, 2008).
- Why do wages differ so much across industries and occupations? (Clark, 2003 and Pischke, 2010).
- How much inequality should there be in a society? (Clark and D'Ambrosio, 2015).
- How harmful is inflation compared to unemployment? (Di Tella *et al.*, 2001).
- How can we value public goods, such as green spaces, pollution and aircraft noise? (Van Praag and Baarsma, 2005; Luechinger, 2009 and Krekel *et al.*, 2016).

My task here will be to look back at what has been done over the past productive 20 years in terms of subjective well-being research. This will cover the three topics mentioned above. But my task is more than that: it is also to set out a kind of wish list of what I would like to see achieved in the next 20 years.

The remainder of the article is organized as follows. Section 2 considers the determinants of well-being, and then Section 3 addresses the consequences. Section 4 discusses some applications of the estimated coefficients in well-being regressions to the labor market and elsewhere, and Section 5 some areas where I think research on well-being may progress in the future. Last, Section 6 concludes.

2. The Correlates of Subjective Well-Being

This is in a way the most obvious question that can be asked regarding subjective well-being: What does its distribution look like in terms of observable characteristics? These characteristics can be at both the individual level and at a more

²Clark and Oswald (1994), Oswald (1997), Easterlin (2001) and Ferrer-i-Carbonell and Frijters (2004).

³Clark and Oswald (1996) and Blanchflower and Oswald (2004).

aggregate level.⁴ We then turn to the essential policy questions of social comparisons and adaptation.

2.1. Individual Characteristics

The individual correlates of various measures of subjective well-being were the subject of some of the very first work in Economics: the role of income and relative income (Hamermesh, 1977), trade unions (Borjas, 1979) and race and sex (Bartel, 1981). The literature since then has grown enormously, and covered a wide variety of attributes of individuals and the jobs that they do.

Perhaps the most widely-studied of all personal characteristics is income. It is by now almost universally accepted that individuals who are observed to have higher levels of income in cross-section data also report higher levels of subjective well-being. The literature is by now far too vast to survey satisfactorily, but some examples using individual-level data are Kahneman and Deaton (2010), using the Cantril ladder as the outcome⁵ and data from the Gallup-Healthways Well-being Index (GHWBI), Luttmer (2005), using life satisfaction as the outcome and data from the National Survey of Families and Households, and Layard *et al.* (2010), using data from the General Social Survey and the German Socio-Economic Panel (SOEP). At the country level, Deaton (2008) uses data from the Gallup World Poll to show that richer countries have higher average scores on the Cantril ladder. These kinds of empirical analysis introduce income in logarithmic form: this corresponds to the subjective well-being effect of an additional dollar of income being larger for those with lower incomes than for those with higher incomes.

After income, one of the individual attributes that has attracted the most attention is unemployment. Here again, the evidence is pretty much unanimous: unemployment is associated with lower levels of well-being both in cross-sectional data (comparing one individual who is unemployed to another who is in employment) and in panel data (looking at the same individual over time as their laborforce status changes between working and unemployment). This kind of finding was an early staple in psychology (Jahoda, 1982), and has made repeated appearances in the economic literature (Clark and Oswald, 1994; Winkelmann and Winkelmann, 1998 and Grün *et al.*, 2010: see also Layard *et al.*, 2012)

Other work has considered characteristics that are of interest in a descriptive sense, but which are not themselves amenable to policy interventions, such as age, sex, ethnicity and country of birth. But it is not because these characteristics cannot be changed that we do not care about them: a good example is the great attention that has been paid to the gender and race patterns of pay in the labor market.

⁴The literature on the individual-level determinants (and some of the aggregate ones) is by now huge, and I will only be able to mention a small number of references here. Longer reviews appear in Dolan *et al.* (2008) and Layard *et al.* (2012).

⁵In which respondents are asked the following question: "*Please imagine a ladder with steps numbered from zero at the bottom to 10 at the top. The top of the ladder represents the best possible life for you and the bottom of the ladder represents the worst possible life for you. On which step of the ladder would you say you personally feel you stand at this time?*"

With respect to age, one extremely stable finding is that of a U-shaped relationship between subjective well-being and age (two examples are Blanchflower and Oswald, 2008 and Cheng *et al.*, 2017). While this finding is widespread, there is no current agreement on why well-being should behave in this way. Sociallydetermined explanations have been called into question by research showing the same U-shape in Great Apes (Weiss *et al.*, 2012).

There are notable sex differences in subjective well-being. In particular, women often report higher values on cognitive-evaluate measures than do men, but also higher stress scores (Nolen-Hoeksema and Rusting, 1999). The analysis of GHWBI data in Kahneman and Deaton (2010) shows that women report more positive affect and higher scores on the Cantril ladder, but also more negative affect and greater stress (see their Table 1). There is debate over the explanation of women's higher satisfaction scores. In the labor market, Clark (1997) considered the potential roles of observable characteristics (such as age and education), selection (of more satisfied women) into the labor market, and different work values in British Household Panel Study (BHPS) data, without being able to fully explain the observed job satisfaction premium in favor of women. He finally suggests a possible role for expectations. If outcomes are related to expectations, and women have lower expectations on the labor market than do men, then any given job outcome will be on average evaluated more positively by women than by men. Suggestive evidence along these lines comes from there being no gender difference in job satisfaction for younger women, women with degrees, women with professional working mothers, and those working in maledominated workplaces. All four of these groups are likely to have higher expectations regarding the labor market. Sousa-Poza and Sousa-Poza (2003) noted that the gender job satisfaction gap in BHPS data has been shrinking over time, which is consistent with women's rising expectations, and the most recent contribution using the BHPS finds that it has disappeared completely (Green et al., 2018). Stevenson and Wolfers (2009) also find evidence of falling female happiness in a number of countries.

The country ranking of well-being is a subject that has attracted a huge amount of interest, in both academic research and the popular press. Helliwell and Wang (2012) analyse data from the Gallup World Poll and produce a wide variety of country rankings. The results for the Cantril ladder and life satisfaction broadly have OECD countries (and especially Scandinavian countries) towards the top of the world ranking (although both Japan and South Korea lag notably behind other OECD countries in this respect), and poorer African countries towards the bottom. This country ranking is not identical for different subjective well-being measures. Asian countries do rather better on happiness measures, and South American countries are more highly-ranked when the outcome is instead positive affect.

With respect to race, there is an established Black-White well-being gap in the US, which seems to be shrinking over time (Stevenson and Wolfers, 2008). Deaton and Stone (2016) also find a race difference in GHWBI data, with Blacks reporting lower Cantril ladder scores than Whites or Others.⁶

⁶They also underline a significant framing effect, with individuals who were first asked about their satisfaction with the way things are going in the US reporting lower individual well-being scores in the next question. This treatment effect is smaller for Blacks than for Whites, and serves to eliminate the Black-White well-being gap (see Figure 1 in Deaton and Stone, 2016).

^{© 2018} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

Other individual characteristics do change within individuals over time, such as income and labor-market status as discussed above. Without going into an extensive list of these characteristics, we finish this subsection by considering three individual characteristics that have attracted empirical attention: health, family life and education.

The first of these is the most problematic. It is very commonplace to find that health and life satisfaction, for example, are correlated. However, health is very often measured in surveys via a self-report question.⁷ As such, both health and life satisfaction are contemporaneous subjective reports that are potentially contaminated by common mood effects (or any other hidden common factor that affects both reported health and well-being). Research has avoided this pitfall by using more objective measures of health, such as the number of health problems that the individual experiences (with respect to sight, hearing, chest, heart etc.), disability, the number of doctor visits over the past year, and the number of days spent in hospital over the past year. The results continue to show a positive correlation between health and subjective well-being.

Marital status is systematically correlated with well-being. The married declare higher levels of satisfaction and happiness than do the single, or the divorced, separated or widowed. Marriage is of course an individual choice, and finding that the married are happier than the single could either show that marriage brings about well-being, or that it is "happy types" who are more likely to become married in the first place. An analogous argument can be made for individuals who divorce. The use of panel data can help us to evaluate selection into marital status as a function of subjective well-being. Stutzer and Frey (2006) analyze this question using SOEP data, and find that the well-being differential associated with marriage is indeed more due to those who were happier when single being more likely to become married in the future.

Perhaps even more than marital status itself, the relationship between children and well-being has produced a wide variety of diverse findings. Cetre *et al.* (2016) consider data from the Gallup World Poll, the European Social Survey (ESS), and the SOEP. They conclude that the relationship between children and happiness differs sharply across countries and populations. In particular, this correlation depends on income, and turns from negative to positive at a GDP level of around 20 000 US Dollars, and on age, being more positive for older adults. Using the panel SOEP data, and following the same kind of approach as used by Stutzer and Frey (2006) for marital status above, it is shown that (prior to any fertility) individuals who will later become parents report higher levels of life satisfaction than do those who will never have children.

The existing work on education has produced a mixed bag of findings. How can this be the case when one of the staple findings in labor economics is that education produces greater earnings (and income is positively correlated with subjective well-being)? One first point is that education may not only raise incomes, but also raise expectations in general about the kind of job or life that the individual

⁷For example, in the BHPS the subjective health question is "*Please think back over the last 12* months about how your health has been. Compared to people of your own age, would you say that your health has on the whole been," with response categories of Excellent, Good, Fair, Poor and Very Poor.

^{© 2018} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

believes that they should have:⁸ in this case the net effect of education on happiness will depend on the rise in outcomes relative to that in expectations. Education is also an individual choice variable, and it is entirely possible that happier people (say) decide to obtain more education. The ideal experiment here would be to randomly allocate education across individuals, and then look at their subjective well-being: this would reveal the causal effect of education.⁹ Various reforms have indeed changed access to education for certain groups: for example, the GI Bill in the US and the raising of the compulsory minimum school leaving age in a number of countries. Oreopoulos and Salvanes (2011) find that a later school-leaving age increased both the income and the well-being of those affected by the change.¹⁰

2.2. Aggregate Characteristics

As well as their own individual characteristics, subjective well-being is systematically linked to the aggregate characteristics of the society in which they live. Perhaps the first of these latter to be systematically investigated was GDP per capita. While Easterlin (1974) found only a weak relationship between well-being and per capita income (based only on a small number of countries), this conclusion has been strengthened in more recent data. In Table 3.1 of Layard *et al.* (2012), GDP per capita explains 65 percent of the cross-country variation in the Cantril ladder in the Gallup World Poll.

The discussion in Section 2.1 above noted that individual unemployment was associated with sharply lower levels of well-being. But what about the aggregate unemployment rate? This is typically found also to be negatively correlated with happiness: one well-known contribution in this respect is Di Tella et al. (2001), who use information from the Eurobarometer. Di Tella et al. not only correlate country-level life satisfaction with the unemployment rate, they also consider the country inflation rate at the same time. The inflation rate is of course a purely aggregate phenomenon. The comparison of the estimated coefficients on aggregate unemployment and aggregate inflation allows us to calculate a trade-off between the two: How much more unemployment would individuals be ready to accept (in the sense of remaining at the same level of life satisfaction) in order to reduce inflation by one percentage point?¹¹ One standard measure of an economy's health is the misery or Okun index, defined as the sum of the unemployment and inflation rates. The results in Di Tella et al. reveal that this assumption of equal weighting is not borne out by the data. In their life-satisfaction regressions, the estimated coefficient on unemployment is over twice as large as that on

⁸Clark *et al.* (2015b) analyse Japanese data and find that education is associated with higher levels of happiness, but also with higher scores on a question asking individuals how happy they believe that they should be.

⁹We would also like to do the same thing with respect to marital status, children, income, and so on.

¹⁰Although Clark and Jung (2016) cannot replicate the UK results in their analysis of BHPS data, finding a life satisfaction effect that is either zero or negative.

¹¹This use of subjective well-being regressions to calculate trade-offs is one of the more interesting applications of this new kind of data. It complements existing work that has tried to infer individual preferences via behavior (i.e., revealed preferences) and hypothetical-choice experiments.

inflation: at the country level, one point of unemployment affects subjective wellbeing more than does one point of inflation, which is surely a piece of information that can be used to guide macroeconomic policy. This conclusion is confirmed in a recent update of the analysis using newer Eurobarometer data by Bell *et al.* (2014).

As well as the level of income (in the region, or country or whatever), we may also be interested in its distribution. The question of the relationship between inequality and happiness has by now produced quite a number of papers. This research area is surveyed in Clark and D'Ambrosio (2015). Although it may seem natural to think that individuals dislike inequality, there are a number of qualifications that can be made to this simple statement (regarding inequality as an indicator of my own future opportunities, deserved versus undeserved inequality, perceived versus actual inequality, and the role of individual comparisons to others' incomes, as in Section 2.3 below). Perhaps unsurprisingly in this light, the empirical literature on the link between happiness and inequality has come up with a wide variety of findings: in Table 13.1 of Clark and D'Ambrosio (2015) only around half of the 30 or so listed contributions suggest a negative relationship. There is obviously still much to learn here.

Other work has looked at the effect of the environment, considering climate, green spaces and pollution. There are by now a number of carefully-crafted pieces of work showing that pollution reduces life satisfaction: two examples are Van Praag and Barsma (2005) with respect to aircraft noise, and Luechinger (2009) regarding air pollution. Green spaces and amenities in general raise the wellbeing of those who can access them: see Ferreira and Moro (2010), Ferrer-i-Carbonell and Gowdy (2007), Lora (2008) and Krekel *et al.* (2016). Rehdanz and Maddison (2005) consider the role of climate, and in particular temperature and precipitation. All of these variables are potential policy targets. Public intervention can and has changed the distribution of green spaces and environmental pollution. Although less directly accessible, climate change is one of this century's key challenges, and Rehdanz and Maddison (2005) specifically speculate on how climate change might affect subjective well-being.

The relationship between government in general and subjective well-being has also attracted attention. Research here has considered, amongst others, the size of government in general (Bjørnskov *et al.*, 2007 and Flavin *et al.*, 2014), the quality of government (Helliwell and Huang, 2008), direct democracy (Frey and Stutzer, 2000 and Flèche, 2015), the government's welfare-state position (Pacek and Radcliff, 2008), and its position relative to the progressivity of taxation (Oishi *et al.*, 2014). Research into some institutions such as trade-unionism has revealed correlations with subjective well-being both at the macro (Flavin *et al.*, 2010) and at the individual (union-membership) levels (Bryson *et al.*, 2004).

Some contextual variables result directly from what other people do. We can here consider for example crime, trust, corruption and altruism. Many of these are measured in the World Values Survey: Trust ("In general, do you think that most people can be trusted, or alternatively that you can't be too careful in dealing with people?"), Altruism ("Have you donated money to a charity in the present month?") and Social Support ("If you were in trouble, do you have relatives or friends you can count on to help you whenever you need them?"). Helliwell et al.

^{© 2018} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

(2018) uncover a strong relationship between trust and well-being, suggesting that a movement from the lowest to the highest level of trust is worth one point on the 0-10 life satisfaction scale, which is a large effect. Generosity, as measured by charitable giving, is also correlated with well-being. While it is undeniably pleasant to be surrounded by generous people, there is also intriguing work suggesting that altruism raises the well-being of those who are doing the giving (see Aknin *et al.*, 2013 and Dunn *et al.*, 2008). Last, social support has been shown to be important in World Values Survey data: societies where everyone has someone to rely on are predicted to have life satisfaction scores that are two points higher than a society where no-one has anyone to rely on.

Of course, not all the behaviors that others engage in are positive: one obvious example is crime. The local quarterly crime rate in the local Police Force Area has been shown to reduce individual mental health in BHPS data (Dustmann and Fasani, 2016).¹² One salient point to be raised here is that the largest effect of crime on well-being may not transit through actually being a victim of crime, but rather via the fear of becoming a future victim of crime (see Moore and Shepherd, 2007).¹³

One recent example of empirical analysis that attempts to introduce a number of these different types of aggregate variables at the same time is provided in Table 3.1 of Layard *et al.* (2012). This analyzes the relationship between countrylevel life satisfaction and measures of social support, freedom, corruption, healthy life expectancy, education and income, all at the aggregate level, and the proportion of individuals who are separated, divorced or widowed. The data here is from the Gallup World Poll, and the well-being measures are the Cantril ladder, positive affect and negative affect. The results for the Cantril ladder show significant relationships with income, health and social support (of around the same size) and smaller, but still significant, relationships with freedom and corruption. Marital status and education are not significant in these regressions. The positive and negative affect analysis reveals a role for social support, freedom and corruption: GDP per capita is not significant here (see also Kahneman and Deaton (2010) for a stronger relationship, using individual-level Gallup data, between income and the Cantril ladder than between income and affect).

The past two sub-sections have underlined that both individual and societal characteristics are correlated with subjective well-being. Many will argue that the ultimate aim of studying individual well-being is to help design policies (at the individual or more aggregate level) that will produce greater well-being. To what extent can we use the findings above to guide well-being policy?

There are in particular two phenomena which may prevent these correlations from being a useful guide to societal welfare maximization. The first of these is adaptation: a certain individual or societal characteristic may be welcome in the short run, but in the longer run we become used to the goods or the state in question. Some of the well-being effects outlined above may then be only temporary

¹²See also Powdthavee (2005).

¹³Of course, the same argument can be made about other life domains as well. While we have highlighted the correlation between being unemployed and subjective well-being, the fear of future unemployment (commonly measured as job insecurity) may be just as important. Equally, the well-being consequences of low income may be paralleled by negative effects from financial insecurity.

in nature. Second, some behaviors may raise the well-being of the individual concerned, but reduce the well-being of others. A very obvious example is crime, but we will argue below that this phenomenon may in fact be widespread. The positive effects of my behavior on my own well-being may be entirely wiped out by negative effects on others' well-being. Again, looking at the correlation between my own characteristics and my well-being only will be a poor guide to policy for societal well-being. Man is not an island, and what other people do is essential for individual well-being (see Clark, 2016c).

2.3. Social Comparisons

Sections 2.1 and 2.2 considered individual and aggregate characteristics respectively. Some of the characteristics are both individual and aggregate, with the aggregate figure (the regional unemployment rate, country GDP per capita) being the sum of the individual-level values. We may wonder how these two levels interact. By far the most-often analyzed characteristic in this respect has been income. While richer countries are happier than poorer countries as a rule, how is the effect of one's own income moderated by income at a more aggregate level? In other words, is a real annual income of \$40 000 per year just as satisfying in Korea as it would be in a country with significantly lower GDP per capita?

The subject of relative income, or income comparisons, was considered at length in Clark *et al.* (2008a). The broad idea here is that subjective well-being (W) is affected positively by own income (Y), but negatively by the income of some reference group (Y^*) .¹⁴

$$W = f(Y, Y^*)$$

In equation (1), individuals are happier when they earn more, but less happy as others earn more. This is most often suggested to reflect social status, envy or some similar phenomenon. Note that Section 2.1 discussed the relationship between individual income and subjective well-being. But it is not enough to know this relationship to make good policy: we need to know how the individual's income affects the well-being of others as well.

There is by now a very substantial literature that has used a variety of approaches to show that individuals do have well-being functions of this type. In terms of observed behavior, there is large literature on conspicuous consumption (whereby individuals gain status from having more of a particular good than do others). Conspicuous consumption is by nature comparative: you can only be conspicuous relative to what others are doing. A couple of empirical examples are Bloch *et al.* (2004) and Brown *et al.* (2011).

Alternatively, it can be shown directly that measures of individual well-being are negatively correlated with measures of Y^* . For example, Clark and Oswald

¹⁴To carry out empirical analysis here, we do need to know who is in the reference group. The literature has used a number of different plausible definitions: those who have the same characteristics (for example, age, sex and education) as me; others in the same household; friends; neighbours; work colleagues; or even myself in the past. For the last definition, the comparison income level to which I would compare my current income is the income that I have myself earned in the past.

(1996) analyze BHPS data and show that the job satisfaction of British employees rises with the individual's own income, but falls with the income of the peer group (other people with the same job and demographic characteristics) rises. Clark (1996a) carries out the same type of analysis, but with the comparison being the partner's income. Luttmer (2005) is a careful attempt using American data to measure comparisons to those who live close to the individual; see also Ferrer-i-Carbonell (2005) using German data. Bellet (2017) is related here, as it considers a measure of wealth, housing, rather than income. Bellet uses American Housing Survey data to show that housing satisfaction is positively correlated with the size of one's own house, but negatively correlated with the size of the houses that have been built in the county since the household moved in: this in particular applies to the average size of all houses bigger than the household's own house, and the average size of the biggest ten percent of houses (see his Table 1).

Income comparisons have also been analyzed using questions on how much money you would need to reach a certain level of well-being (the Leyden "Welfare Function of Income": see Van de Stadt *et al.*, 1985), experimental data (Zizzo and Oswald, 2001 and Abbink *et al.*, 2009), questions on preferences over hypothetical outcomes (Alpizar *et al.*, 2005; Solnick and Hemenway, 2005), and neurological evidence from brain activity in situations where social comparisons are salient (Fließbach *et al.*, 2007).

The conclusion from this work is that well-being is at least partly comparative in income. This is one explanation of the Easterlin Paradox (Easterlin, 1974). If we consider Y^* to be the country's GDP per capita, then at any point in time richer individuals will be happier than poorer individuals (holding Y^* constant in equation (1), individual well-being rises in income). But as everyone becomes richer, both Y and Y^* rise, and their effects potentially cancel each other out. Higher GDP per capita will then not be accompanied by greater happiness. Easterlin (2017) is a recent discussion.

While income comparisons have attracted the most attention in this literature, the variable Y in equation (1) can refer to outcomes in many different life domains. In Economics, the relationship between unemployment and individual well-being has figured regularly. The question of social comparisons over unemployment was addressed in Clark (2003) using BHPS data. The well-being effect of unemployment is context-dependent, with own unemployment having a smaller effect on well-being in higher-unemployment regions and in higherunemployment households. The surprising finding that unemployment matters less in well-being terms when it is shared with others is in line with research on suicide and para-suicide rates by the unemployed, which are highest in lowunemployment regions (Platt *et al.*, 1992).

Research on individual health status has also shown evidence of comparisons. Blanchflower *et al.* (2009) find that self-reports of being overweight rise with own Body Mass Index (BMI), but fall with the average BMI of the reference group (defined as country*age group*sex cells) in Eurobarometer data. Carrieri and de Paola (2012) present analogous findings for well-being and height in Italian cross-section data. For a given level of own height, those who are taller than the average of their peer group (at the region*age*sex level) report higher levels of well-being. Clark and Etilé (2011) also consider BMI, but with the comparison

^{© 2018} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

being to the partner. The analysis of SOEP data reveals that (as is standard) at least after a certain point own BMI and well-being are negatively related. However, the strength of this relationship is moderated by partner's BMI. In particular, the negative effect of own obesity on well-being is smaller when one's partner is either overweight or obese.

Similar kinds of social-context effects on well-being have been uncovered in the domains of education (Nikolaev, 2016) and religion (Clark and Lelkes, 2009). In general, it seems that there is much more work that can be done to uncover the effects of social comparisons in the determination of subjective well-being.

2.4. Adaptation

We now turn to adaptation. The correlations described in Sections 2.1 and 2.2. above were contemporaneous. But for policy purposes, we need to know if these effects will fade away in the longer-run. A recent survey of this area is Clark (2016b). Note that adaptation is just a particular kind of comparison, as in equation (1), where the reference group is now not other people but rather the individual himself or herself in the past.

In terms of comparisons to income, today's income is evaluated in terms of how much I used to earn in the past. My own past good fortune then reduces my current well-being. Clark (1999) used BHPS panel data to define Y^* as the income of the same individual in the same job one year earlier, and found that own past earnings are indeed negatively correlated with current job satisfaction. Similar results have been obtained in SOEP panel data: see for example Grund and Sliwka (2007), Di Tella *et al.* (2010) and Vendrik (2013). The finding of a negative relationship between past income and current subjective well-being, conditional on the level of own income, is synonymous with individuals adapting to higher incomes.

As for the discussion of social comparisons in Section 2.3, there are a number of "non-happiness" ways of looking at adaptation to income. Work on the Leyden Welfare Function of Income has shown that individuals who earned more in the past say that they require a higher level of income today in order to be satisfied.¹⁵ It is also possible to use information on observed behavior to look for evidence of income adaptation. This has been carried out with experiments on revealed preferences over hypothetical income profiles (Loewenstein and Sicherman, 1991 and Frank and Hutchens, 1993), and the analysis of survey data on labour supply (Hotz *et al.*, 1988).

If individuals adapt to income, then the well-being response to higher income will be only temporary, as in Figure 1 below.

In Figure 1, a permanent rise in income at the end of year 2 produces a contemporaneous rise in well-being. But this effect does not last, with well-being progressively dropping back down to its pre income-rise level. As in Section 2.3 above, we might wonder whether the same phenomenon of adaptation is found in other life domains. This work has been carried out for the labour market, marital status, children, and health in particular. The analysis of adaptation to

¹⁵Van Praag (1971) calls this "preference drift:"

^{© 2018} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth



Figure 1. Happiness Adaptation following an Income Shock Source: Clark (2016b).

unemployment, for example, consists in following the same individual in panel data before and after they enter unemployment. Of key interest here is whether the pattern of well-being while the individual remains unemployed shows evidence of adaptation (by first dropping and then bouncing back).

Figure 2 below illustrates the adaptation results for five life events (unemployment, marriage, divorce, birth of child, and widowhood) in BHPS data from Clark and Georgellis (2013). The results here are similar to those in the SOEP (Clark *et al.*, 2008b) and in other panel datasets (as surveyed in Clark, 2016b). Unemployment stands out here as being a status to which individuals do not adapt. On the contrary, there is evidence of fairly complete adaptation to all of marriage, children, divorce and widowhood.

While Figure 2 (and work on other panel datasets besides) concludes for only a temporary well-being impact of marriage, in most of the countries analyzed marriage is preceded by a number of years of being in a couple. When we instead apply the same analysis technique to becoming a couple (i.e. entering a relationship), we instead find only partial or no adaptation in BHPS, SOEP and Australian HILDA data (see Chapter 5 of Clark *et al.*, 2018).

There is still much work to be done in the area of well-being adaptation using large-scale panel datasets. The evidence so far suggests that adaptation is not a universal truth. As can be seen in Figure 2, unemployment is an outlier in terms of the events considered there. But it is not the only one. Some of the work cited in Clark (2016b) concludes as to partial or no adaptation to disability, cosmetic surgery and job insecurity (but full adaptation to moving house and becoming self-employed). In addition, although there is evidence for adaptation to income in general, Clark *et al.* (2016b) use SOEP data to show that individuals do not adapt to entry into poverty (and indeed do not adapt to drops in income in

^{© 2018} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth



Figure 2. Adaptation to life events in the BHPS

Notes: X, Δ and \Box denote significance at the 1 percent, 5 percent and 10 percent levels respectively; the error bars represent the 95 percent confidence intervals. The analysis here concerns people who became unemployed (for example) at time zero and who stay unemployed over the period in question. The change in life satisfaction between times t+1 and t+2 is the average well-being (conditional on other control variables) of individuals who are unemployed at t+1 and remain unemployed at t+2 (*Source*: Clark and Georgellis, 2013).

general). This underlies that there might be different well-being effects according to the direction of the change that we identify in panel data, which is one of the frontlines of research that I will consider in Section 5 below.

3. What do Happy People do?

The literature on the determinants of happiness has been growing fast. But so has that on the consequences of well-being. This literature has is in part been carried out with the validation of subjective well-being data in mind. One common objection to the use of such data is exactly that they are subjective. However, if happier people can be shown to be more likely to have some objective outcome in the future, then there must be some "real" information in the subjective wellbeing scores that different individuals report. The advantage of using subjective data to predict future observable outcomes is that economists have traditionally been more comfortable with objective measures of behavior (what people do) than with subjective evaluations (what people say).

Section 2.1 above asked whether income buys happiness; we can also ask the reverse question of whether happy people earn more. Well-being has been shown to increase productivity (Oswald *et al.*, 2015), and to help predict future earnings (De Neve *et al.*, 2013). In Flèche (2017), teacher satisfaction predicts pupil value-added in Maths: as such, satisfaction can not only affect the individual's own outcomes, but also those of others. Workers with higher well-being scores have been shown to be less likely to quit their jobs (Freeman, 1978 was an early contribution here) and less likely to retire (Clark *et al.*, 2015c).

Individuals with higher well-being scores also have better health outcomes. One famous contribution here is the "Nun Study," in which nuns who wrote more positive short descriptions of their life in their late teens and early twenties were significantly more likely to still be alive 60 years later (Danner *et al.*, 2001). Banks *et al.* (2012) find large-scale equivalent findings relating measures of well-being to survival probabilities six years later in English Longitudinal Survey of Ageing data. In Diener and Chan (2011), happier people are shown to have stronger immune systems, experience less inflammation and cardiovascular disease and suffer from fewer infections. They also suffer from less telomere shortening (see Epel, 2009).

Turning to marriage and the family, as noted in Section 2.1 above, happiness scores help to predict future marriage. Along the same lines, in SOEP data lower levels of life satisfaction predict future divorce or separation (Clark *et al.*, 2008b) and future fertility (Cetre *et al.*, 2016).

So being happy is good for you. And it may also be good for other people around you. Leaving to one side the vexed question of whether happiness is contagious, it has been shown that subjective well-being helps predict social behaviors that others value. As noted in Section 2.2, individuals with higher (previously-elicited) happiness scores are subsequently more generous when deciding how much money to give to others; equally, Thoits and Hewitt (2001) suggest that happier individuals volunteer more. Flavin and Keane (2012) underline a relationship between subjective well-being and political participation. Two recent contributions in this domain are

Liberini *et al.* (2017), who find that life satisfaction predicts political support for the incumbent government in BHPS data (even controlling for income and the change in the financial situation), and Ward (2017), who uses 40 years of Eurobarometer data covering 15 European countries to show that life satisfaction is a stronger predictor of incumbent vote share than are economic growth, unemployment and inflation. Last, De Neve *et al.* (2013) note that happier people are in general more cooperative and more sociable.

The mirror image of this correlation is that those with lower well-being are more likely to engage in activities that hurt others. Layard *et al.* (2014) find that average childhood emotional health over the ages 5 to 16 predicts (negatively) the number of adult crimes between the ages of 16 and 34 in British Cohort Study data.

4. Some Uses of Well-Being Data

The previous two sections have addressed the key questions of what determines happiness, and what does happiness predict. In addition to these central issues, subjective well-being has been exceedingly useful in providing estimates of economic magnitudes that are difficult or impossible to obtain from other sources.

Preferences are not absolute. It is not helpful to say "I like X" or "I like Y very much." In a world of constrained maximisation, we need to know how much you would be willing to give up to have one unit more of X or Y. Were we to be able to observe some cardinal measure of utility, then we could say (for example) that one more hour of leisure is worth some number of utils. But many people don't think of utils as a natural unit of measure.¹⁶

We often refer to trade-offs in order to quantify preferences: how much money would you give up to have that new car (i.e. What is the car's shadow price)? Or how much extra weekly labour income would you require to work one more hour per week (i.e. What is your shadow wage)?

Information on these trade-offs can be obtained from a life-satisfaction equation that is estimated as a function of weekly income (Y), weekly hours of work (h) and a set of other controls, \underline{X} :

(2) Life satisfaction = $\beta_1 Y + \beta_2 h + \theta' X + \varepsilon$

Here one more hour of work per week would be compensated (in the sense of leaving life satisfaction unchanged) by a rise in weekly labor income of $-\beta_{2/}\beta_1$. Clark (1996b) was an early attempt at this kind of calculation on data from the first (1991) wave of the BHPS, producing a reasonable-sounding figure of £8.60 per hour.¹⁷ A far more recent estimation of this type, appealing to an exogenous movement in hours of work due to the reductions in the standard workweek in

¹⁷This is equivalent to around £17.50 per hour in 2017 prices.

¹⁶Although Clark *et al.* (2018) argue that it is becoming more natural to think of life satisfaction on a zero to ten scale as an outcome, so that individuals do have an idea of what one life-satisfaction point represents.

^{© 2018} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

France and Portugal, appears in Lepinteur (2016). The same approach is taken for shorter workweeks in Japan and Korea in Hamermesh *et al.* (2017).

Running an equation such as (2) produces one estimated coefficient on each variable, and so only one value for the trade-off between two determinants of life satisfaction. If we would instead like to predict why some people act differently from others (i.e. to predict behavior) then we require the trade-offs to be different for across groups of individuals. We will likely never have enough data to estimate (2) separately for each individual, but we can separate individuals up into groups. This can either be done ex ante, by running separate regressions for men and women, or for the older and the younger, or by using latent-class estimation techniques, where a number of different regression lines (and so estimated coefficients) are identified endogenously via maximum likelihood (see Clark *et al.*, 2005). The findings from the latter kind of estimation have shown that those with a higher estimated marginal utility of income (a higher value of β_1 in equation (2) above) retire from the labour market later (Clark and Fawaz, 2016) and tend to vote more right-wing (Clark and Etilé, 2013).

The estimated trade-offs in subjective well-being equations can be compared to those obtained by other more traditional means. The estimated income-leisure trade-off seem to be similar in subjective well-being equations and standard labor-supply estimation (Akay *et al.*, 2015). However, Benjamin *et al.* (2014) strike a note of caution, as their estimated trade-offs in life-satisfaction equations do not correspond well to the trade-offs that individuals actually make in observed choices.

Subjective well-being analysis can also inform us about the value of the kinds of aggregate variables discussed in Section 2.2 above. As noted there, Di Tella *et al.* (2001) compared the estimated coefficients on unemployment and inflation. We can also compare the estimated coefficient on income to that on a more aggregate variable to calculate the nuisance value of noise pollution (as in Van Praag and Baarsma, 2005) and air pollution, or the value of access to green spaces, for example.

The estimated coefficients in well-being equations also contribute to the understanding of choice. One initial vexed question on the labor market concerned the extent of voluntary unemployment (in the aftermath of the huge rise in unemployment following the first and second oil-price shocks in the 1970s). If unemployment is chosen over employment, then it should yield a higher level of well-being. Clark and Oswald (1994) compared the well-being of the employed to the unemployed in the first wave of the BHPS and found a large gap between unemployment and employment, which is not consistent with unemployment being for the most part a choice. In the same vein, a very well-known contribution by Blanchflower and Oswald (1998)¹⁸ looked at self-employment versus employment: Is self-employment a choice, or is it rather the last resort of those who cannot find an adequate job? Job- and life-satisfaction data reveal that the self-employed actually have higher levels of well-being than do the employed. This itself poses a new question: Why are we not all self-employed then, if it is preferable to employment? Blanchflower and Oswald hypothesize that some people

¹⁸This is the most-cited paper ever published in the Journal of Labor Economics.

^{© 2018} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

would like to become self-employed, but are not able to obtain the necessary capital to do so (as there is asymmetric information, and banks do not necessarily fund all good projects). As a test, the sample is split into those who inherited money and those who did not, with the argument that the former are less likely to be capital-constrained. The satisfaction differential between the employed and the self-employed is large and significant for the latter group, but zero for the former group. In the same way that the unemployed would prefer employment (but can't find a job), some employees would prefer to be self-employed (but cannot obtain the necessary capital to do so).

The same kind of analysis of allocation on the labor market can be carried out within employees, looking at public- versus private-sector jobs (Luechinger et al., 2008), part-time versus full-time jobs (Booth and Van Ours, 2008), and temporary versus permanent work (Busk et al., 2017). Moving away from simple binary comparisons of this type, we can consider the choice of different occupations and industries. There has been a huge amount of work in labor economics on wage differentials at this level. Yet there is still no agreement on what these represent: Are they compensating for unobserved job characteristics (as in the theory of compensating differentials), or do they reveal that jobs in some occupations and industries are really more attractive than others (so that the wage differentials reflect rents)? One way of distinguishing between these two explanations is to see whether the observed wage differences map onto observed job-satisfaction differences. Any such mapping will be consistent with wage rents, while the lack of a correlation suggests compensating differentials rather. This is what Clark (2003) does using BHPS data. There is no correlation between the industry wage and job satisfaction structures; however, there is a strong relationship between the occupation wage and job satisfaction structures. The proposed interpretation is in terms of tournaments. Tournaments involve moving up job ladders (and changing occupation, but not industry), and the "prize" associated with winning the tournament is a wage rent.

Last, we can consider allocations outside of the labour market, and in particular the attraction of different regions. Oswald and Wu (2011) show that the pattern of life satisfaction across the States of the US closely matches the associated distribution of objective amenities.

These applications of the estimated coefficients in well-being regressions still remain relatively unexplored. Much more work in this area is to be expected.

5. CURRENT AND FUTURE FRONTLINES

Since empirical research in subjective well-being in Economics really started to take off in around 2000, there has been a great deal of replication, contention and confirmation, as is normal in pretty much any empirical science. While some results have now entered into well-being lore (relative effects of income, the Ushaped relationship with age), others are much more in the frontline of current research. In addition to the ongoing research mentioned a number of times above, this last section discusses a number of promising research areas, in no particular order.

^{© 2018} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

A first very general point is that research tends to be rather monolithic in terms of the datasets used, and thus in terms of the populations whose well-being we analyse. It is true that there is cross-country work on the Gallup World Poll and the World Values Survey, but large-scale work using panel data has in the vast majority been reduced to three countries: Australia (HILDA), Germany (SOEP) and the UK (BHPS). This restriction is of course by necessity rather than choice, but we should always be aware of the potential lack of external validity of many of our findings.

Second, almost every single well-being paper of the type described in Section 2 has modelled averages. It is undoubtedly true that we care about average wellbeing in a society, but we probably care about its distribution too: for given average satisfaction, we would prefer the variance of well-being to be lower, as this would imply fewer people with low well-being (and our social-welfare function may put more weight on those in misery than on those with high subjective wellbeing). There are very few contributions in this sense. One recent one (Clark *et al.*, 2016a) underlined that the correlates of the mean and variance of wellbeing may well not be the same. While there has been a lot of work on the potential lack of a relationship between GDP growth and average satisfaction or happiness in a country, GDP growth does seem to be systematically correlated with a lower variance in well-being.

Another way of broadening the area of empirical inquiry would be to take quantile effects into account (as in Binder and Coad, 2011 and 2015). A variable that is overall uncorrelated with well-being in a regression may instead exhibit a strong correlation at some point of the well-being distribution, making it potentially useful for policy purposes.

Research has also been concentrated on the adult determinants of adult subjective well-being. There are at least two possible extensions here. One is to consider the distal (childhood and family) correlates of adult well-being (as in Layard *et al.*, 2014, and Clark *et al.*, 2018). The other is to consider childhood well-being as an outcome in its own right. Partly because the childhood we live is part of what makes a good life, and partly because childhood well-being is such a strong predictor of well-being throughout adulthood. The concentration on family and childhood here requires birth-cohort data for the measurement of family background, childhood outcomes, and well-being throughout life.

There has not been enough work on the effect of exogenous movements in explanatory variables on well-being. Of course, a number of key variables are exogenous by construction (age, ethnicity and sex) and others seem to pretty much act is if they were (unemployment). Exogenous movements in some others are quite easy to imagine, such as lottery winnings for income (Apouey and Clark, 2015 and Gardner and Oswald, 2007), or the minimum school-leaving age for education (Clark and Jung, 2016 and Oreopoulos and Salvanes, 2011). However, instruments for other variables are rather more difficult to think of, such as marital status, and children.¹⁹

¹⁹Of course some events are exogenous by nature, and there have been a number of analyses of the well-being consequences of events such as natural disasters (Rehdanz *et al.*, 2015) and terrorism (Frey *et al.*, 2009; Metcalfe *et al.*, 2011 and Clark *et al.*, 2017).

^{© 2018} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

Panel data has greatly advanced our understanding of the causes and consequences of subjective well-being. One obvious contribution has been the ability to control for individual fixed effects, thereby conditioning out any unobserved individual variables that do not vary over time and might be correlated with wellbeing. One obvious candidate here is reporting style, whereby some individuals tend to use lower numbers to reflect the same latent feelings than do others. Panel data also allows us to correlate current well-being to past values of the explanatory variables. This was behind the analysis of adaptation described in Section 2.4; another application is to consider the role of past completed events on current well-being, such as the scarring effect of past unemployment (Clark *et al.*, 2001 and Knabe and Rätzel, 2011). To date this kind of dynamic analysis has only been quite simple. Other more complicated analyses are possible, including the use of lagged dependent variables and the introduction of different time profiles of the explanatory variables (such as the order of different past poverty spells in Clark *et al.*, 2015a).

The use of panel data also allows us to consider potential asymmetries in reactions to positive and negative movements. The estimation of fixed-effects models works via comparing intra-individual changes in well-being to the intra-individual changes in the explanatory variables. This estimation is symmetric, in the sense that the same relationship is assumed to hold when the explanatory variable is above the individual-level mean and below the individual-level mean:

(3)
$$WB_{it} - \overline{WB}_i = \beta(x_{it} - \overline{x}_i) + \varepsilon_{it}$$

In the case of income, a jump of 1000 Euros above the individual-level mean induces a change in life satisfaction of the same absolute size (we here assume that the dependent variable is life satisfaction and the explanatory variable, x, is income) as does a drop of 1000 Euros below the individual-level mean: β . But it is far from clear that this is the case, and loss-aversion would suggest that falls in income produce well-being movements that are larger in size than those from an equivalent rise in income. Some empirical evidence along these lines using SOEP data appears in D'Ambrosio and Frick (2012) and Boyce *et al.* (2013).

This asymmetry is not limited to income, or even to cardinal explanatory variables. Considering a binary explanatory variable, labor-force status say, entry into unemployment from employment is assumed to produce the same movement in life satisfaction in absolute value as the inverse switch when the individual exits unemployment to start a new job. In general, much remains to be explored regarding the role of the time profiles of explanatory variables and current subjective well-being.

The next two research areas concern the exploration of more unusual data in the context of subjective well-being. There has been only very little work on happiness and brain activity (two examples are Urry *et al.*, 2004 and Fliessbach *et al.* 2007). Neuro evidence is very valuable here not only for what it tells us about the sources of well-being, but also because it arguably avoids any bias in well-being scores due to the individual reporting function. The well-being measures here come straight from the horse's mouth, as it were.

^{© 2018} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

There is great interest in the role of genetics. Are we pre-determined by our genes, including in terms of our subjective well-being? While recent research has identified genes that are correlated with subjective well-being (De Neve *et al.*, 2012) this does not mean that happiness is pre-determined. Research has underlined the importance of gene-environment interactions. For simplicity, imagine that there are less-favorable and more-favorable genotypes. Individuals with a more-favorable genotype are not susceptible to low well-being; those with the less-favorable genotype are susceptible to low well-being, but only if they experience negative events. Empirical analysis will reveal a correlation between genotype and subjective well-being. But this is not an individual fixed effect: it rather reflects an interaction between genes and the events that befall the individual. In this simple illustration, policy that avoided the occurrence of negative events (perhaps unemployment or poverty) would prevent the less-favorable genotype from playing a role, effectively shutting off any observed relationship between genes and happiness.²⁰

Last, the rise of subjective well-being across the social sciences has been accompanied by an increase in the availability of ways of measuring it. Which way of measuring well-being is the best? There is no simple way of answering this question, but research has shown that life satisfaction, measures of recent positive and negative emotions (or "affect") and eudaimonic measures (that reflect notions of mastery, relations with others, self-acceptance and purpose) are reasonably similar, especially in terms of how they are related to a standard set of explanatory variables (Clark, 2016a). But they are not the same. It is important to distinguish between affect (how you feel right at this moment: your emotions) and cognitive/evaluative measures such as life satisfaction, which refer to your evaluation of your life overall. We might expect the latter to be some kind of sum of the former, but this does not seem to be the case. This raises the very difficult question of which of affect or life satisfaction we should have as a goal as policymakers. What matters more: life as you live it day-to-day or moment-to-moment, or life as you remember it? Would you prefer to live a good life, or to remember having lived a good life?

There is still a great deal that we do not know about these different measures. For example, the work on social comparisons and adaptation has almost exclusively appealed to information on life satisfaction. But would the same phenomena be revealed were we to consider the emotion of sadness, or Eudaimonia?

Beyond these correlations, future work may appeal to the ability of measures of subjective well-being to predict behavior, as discussed in Section 3 above. There is however a dearth of research of this type comparing a number of different well-being measures. The fascinating work in Benjamin *et al.* (2012) shows that own happiness is not the only factor that predicts individual choice (in hypothetical-choice experiments): there are independent effects of control, purpose, fun and social status. It is worth noting that own happiness is the largest single predictor of choice by far, and that the addition of these non-happiness terms does not improve the predictive power of the equation that much. The search for the "best" measure of well-being is likely to continue for some time.

²⁰See Pluess (2015) for a recent survey of this fascinating area.

^{© 2018} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

6. CONCLUSION

The remarkable rise of subjective well-being across the social sciences, and in particular Economics, seems to continue apace. Once considered as an intellectual curiosity at best, and most definitely not part of Economics, it has seemingly started to enter the mainstream. Happiness articles appear in the top-ranked journals, and are amongst the best-cited articles published by those journals. Just over 15 years ago, a review article asked "What can economists learn from happiness research?" (Frey and Stutzer, 2002). That article is currently the sixth most-cited article ever published in the *Journal of Economic Literature*: the answer would then appear to be "a very great deal." The past four decades of happiness research have been inventive, and to my mind have brought social sciences closer together. The coming decades have every chance of being just as exciting.

References

- Abbink, K., D. Masclet, and M. Van Veelen, "Reference Point Effects in Antisocial Preferences," CBESS Discussion Paper 09-03, 2009.
- Akay, A., O. Bargain, and X. Jara Tamayo, "Back to Bentham: should we? Large scale comparison of decision and experience utility," Aix-Marseille Université, mimeo, 2015.
 Aknin, L., C. Barrington-Leigh, E. Dunn, J. Helliwell, J. Burns, R. Biswas-Diener, I. Kemeza, P.
- Aknin, L., C. Barrington-Leigh, E. Dunn, J. Helliwell, J. Burns, R. Biswas-Diener, I. Kemeza, P. Nyende, C. Ashton-James, and M. Norton, "Prosocial Spending and Well-Being: Cross-Cultural Evidence for a Psychological Universal," *Journal of Personality and Social Psychology*, 104, 635–52, 2013.
- Alpizar, F., F. Carlsson, and O. Johansson-Stenman, "How Much do we Care about Absolute Versus Relative Income and Consumption?," *Journal of Economic Behavior and Organization*, 56, 405–21, 2005.
- Apouey, B. and A. E. Clark, "Winning Big but Feeling No Better? The Effect of Lottery Prizes on Physical and Mental Health," *Health Economics*, 24, 516–38, 2015.
- Banks, J., J. Nazroo, and A. Steptoe, eds., The dynamics of ageing: Evidence from the English Longitudinal Study of Ageing 2002-10 (Wave 5), Institute for Fiscal Studies, London, 2012.
- Bartel, A. P., "Race Differences in Job Satisfaction: A Reappraisal," *Journal of Human Resources*, 16, 295–303, 1981.
- Bell, D., D. G. Blanchflower, A. Montagnoli, and M. Moro, "The Happiness Tradeoff Between Unemployment and Inflation," *Journal of Money Credit and Banking*, 46, 117–41, 2014.
- Bellet, C., "The Paradox of the Joneses: Superstar Houses and Mortgage Frenzy in American Suburbs," CEP Discussion Paper No.1462, 2017.
- Benjamin, D., O. Heffetz, M. Kimball, and A. Rees-Jones, "What Do You Think Would Make You Happier? What Do You Think You Would Choose?," *American Economic Review*, 102, 2083–110, 2012.

——, "Can Marginal Rates of Substitution Be Inferred From Happiness Data? Evidence from Residency Choices," *American Economic Review*, 104, 3498–528, 2014.

Binder, M. and A. Coad, "From Average Joe's happiness to Miserable Jane and Cheerful John: Using quantile regressions to analyze the full subjective well-being distribution," *Journal of Economic Behavior and Organization*, 79, 275–90, 2011.

——, "Heterogeneity in the Relationship Between Unemployment and Subjective Wellbeing: A Quantile Approach," *Economica*, 82, 865–91, 2015.

Bjørnskov, C., A. Dreher, and J. Fischer, "The Bigger the Better? Evidence of the Effect of Government Size on Life Satisfaction around the World," *Public Choice*, 130, 267–92, 2007.

- Blanchflower, D. G. and A. J. Oswald, "What Makes an Entrepreneur?," *Journal of Labor Economics*, 16, 26–60, 1998.
- ——, "Well-Being over Time in Britain and the USA," *Journal of Public Economics*, 88, 1359–86, Blanchflower, D. G. and A. J. Oswald, 2004.

———, "Is Wellbeing U-shaped over the Life Cycle?," Social Science and Medicine, 66, 1733-49, 2008. Blanchflower, D. G., A. J. Oswald, and B. Van Landeghem, "Imitative Obesity and Relative Utility," Journal of the European Economic Association, 7, 528–38, 2009.

Bloch, F., V. Rao, and S. Desai, "Wedding Celebrations as Conspicuous Consumption: Signaling Social Status in Rural India," *Journal of Human Resources*, 39, 675–95, 2004.

- Booth, A. L. and J. Van Ours, "Job Satisfaction and Family Happiness in Britain: The Part-Time Work Puzzle," *Economic Journal*, 118, F77–F99, 2008.
- Borjas, G. J., "Job Satisfaction, Wages and Unions," Journal of Human Resources, 14, 21-40, 1979.
- Boyce, C., A. Wood, J. Banks, J., A. E. Clark, and G. Brown, "Money, Well-Being, and Loss Aversion: Does an Income Loss Have a Greater Effect on Well-Being Than an Equivalent Income Gain?," *Psychological Science*, 24, 2557–62, 2013.
- Brown, P., E. Bulte, and X. Zhang, "Positional Spending and Status Seeking in Rural China," *Journal of Development Economics*, 96, 139–149, 2011.
- Busk, H., C. Dauth, and E. Jahn, "Do Changes in Regulation Affect Temporary Agency Workers' Job Satisfaction?," *Industrial Relations*, 56, 514-44, 2017.
- Bryson, A., L. Cappellari, and C. Lucifora, "Does Union Membership Really Reduce Job Satisfaction?," British Journal of Industrial Relations, 42, 439–59, 2004.
- Carrieri, V. and M. De Paola, "Height and subjective well-being in Italy," *Economics & Human Biology*, 10, 289–98, 2012.
- Cetre, S., A. E. Clark, and C. Senik, "Happy People Have Children: Choice and Self-Selection into Parenthood," *European Journal of Population*, 32, 445–73, 2016.
- Cheng, T., A. J. Oswald, and N. Powdthavee, "Longitudinal Evidence for a Midlife Nadir in Human Well-being: Results from Four Data Sets," *Economic Journal*, 127, 126–42, 2017.
- Clark, A. E., "L'utilité est-elle relative? Analyse à l'aide de données sur les ménages," Economie et Prévision, 121, 151–64, 1996a.
 - , "Job Satisfaction in Britain," British Journal of Industrial Relations, 34, 189-217, 1996b.
 - —, "Are Wages Habit-Forming? Evidence from Micro Data," *Journal of Economic Behavior and Organization*, 39, 179–200, 1999.
 - , "Looking for Labour Market Rents With Subjective Data," PSE, mimeo, 2003.
 - ——, "SWB as a Measure of Individual Well-Being," in M. Adler and M. Fleurbaey, eds, *Oxford Handbook of Well-Being and Public Policy*, 518–52, Oxford University Press, Oxford, 2016a.
 - ——, "Adaptation and the Easterlin Paradox, in T. Tachibanaki, ed., Advances in Happiness Research: A Comparative Perspective, 75–94, Springer, New York, 2016b.
 - —, "Happiness, Habits and High Rank: Comparisons in Economic and Social Life," in S. Bartolini, E. Bilancini, L. Bruni, and P.-L. Porta, eds, *Policies for Happiness*, 62–94, Oxford University Press, Oxford, 2016c.
- Clark, A. E., E. Diener, Y. Georgellis, and R. Lucas, "Lags and Leads in Life Satisfaction: A Test of the Baseline Hypothesis," *Economic Journal*, 118, F222–F243, 2008b.
- Clark, A. E. and C. D'Ambrosio, "Attitudes to Income Inequality: Experimental and Survey Evidence," in A. Atkinson and F. Bourguignon, eds, *Handbook of Income Distribution Volume* 2A, 1147–1208, Elsevier, Amsterdam, 2015.
- Clark, A. E., C. D'Ambrosio, and S. Ghislandi, "Poverty Profiles and Well-Being: Panel Evidence from Germany," *Research on Economic Inequality*, 23, 1–22, 2015a.
- ——, "Adaptation to Poverty in Long-Run Panel Data," *Review of Economics and Statistics*, 98, 591–600, 2016b.
- Clark, A. E., O. Doyle, and E. Stancanelli, "The Impact of Terrorism on Well-being: Evidence from the Boston Marathon Bombing," Geary Institute, University College Dublin, Working Paper No. 201708, 2017.
- Clark, A. E. and F. Etilé, "Happy House: Spousal Weight and Individual Well-Being," Journal of Health Economics, 30, 1124–136, 2011.
- ——, "Values, Votes and Slopes. Political Behaviour and the Marginal Utility of Income," PSE, mimeo, 2013.
- Clark, A. E., F. Etilé, F. Postel-Vinay, C. Senik, and K. van der Straeten, "Heterogeneity in Reported Well-Being: Evidence from Twelve European Countries," *Economic Journal*, 115, C118–C132, 2005.
- Clark, A. E. and Fawaz, Y., "Retirement and the Marginal Utility of Income," PSE, mimeo, 2016.
- Clark, A. E., S. Flèche, R. Layard, N. Powdthavee, and G. Ward, *The Origins of Happiness*, Princeton University Press, 2018.
- Clark, A. E., S. Flèche, and C. Senik, "Economic Growth Evens-Out Happiness: Evidence from Six Surveys," *Review of Income and Wealth*, 62, 405–19, 2016a.
- Clark, A. E., P. Frijters, and M. Shields, "Relative Income, Happiness and Utility: An Explanation for the Easterlin Paradox and Other Puzzles," *Journal of Economic Literature*, 46, 95–144, 2008a.
- Clark, A. E. and Y. Georgellis, "Back to Baseline in Britain: Adaptation in the BHPS," *Economica*, 80, 496–512, 2013.
- Clark, A. E., Y. Georgellis, and P. Sanfey, "Scarring: the Psychological Impact of Past Unemployment," *Economica*, 68, 221–41, 2001.
- Clark, A. E. and S. Jung, "Compulsory Education and Life Satisfaction," PSE, mimeo, 2016.

- Clark, A. E., A. Kamesaka, and T. Tamura, "Rising Aspirations Dampen Satisfaction," *Education Economics*, 23, 515–31, 2015b.
- Clark, A. E. and O. Lelkes, "Let us Pray: Religious Interactions in Life Satisfaction," PSE, Discussion Paper 2009-01, 2009.
- Clark, A. E., K. Mavromaras, and Z. Wei, "Happy to Stay: Job Satisfaction and Retirement," Flinders University, mimeo, 2015c.
- Clark, A. E. and A. J. Oswald, "Unhappiness and Unemployment," *Economic Journal*, 104, 648–59, 1994.

, "Satisfaction and Comparison Income," Journal of Public Economics, 61, 359-81, 1996.

- D'Ambrosio, C. and J. Frick, "Individual Well-Being in a Dynamic Perspective," *Economica*, 79, 284– 302, 2012.
- Danner, D., D. Snowdon, and W. Friesen, "Positive Emotions in Early Life and Longevity: Findings from the Nun Study," *Journal of Personality and Social Psychology*, 80, 804–13, 2001.
- De Neve, J.-E., N. Christakis, J. Fowler, and B. Frey, "Genes, Economics, and Happiness," Journal of Neuroscience, Psychology, and Economics, 5, 193–211, 2012.
- De Neve, J.-E., E. Diener, L. Tay, and C. Xuereb, "The Objective Benefits of Subjective Well-Being," in J. Helliwell, R. Layard and J. Sachs, eds, *World Happiness Report 2013*, 58–89, Columbia Earth Institute, New York, 2013.
- Deaton, A., "Income, Health and Well-Being around the World: Evidence from the Gallup World Poll," *Journal of Economic Perspectives*, 22, 53–72, 2008.
- Deaton, A. and A. Stone, "Understanding context effects for a measure of life evaluation: how responses matter," Oxford Economic Papers, 68, 861–70, 2016.
- Diener, E. and M. Chan, "Happy People Live Longer: Subjective Well-Being Contributes to Health and Longevity," *Applied Psychology: Health And Well-Being*, 3, 1–43, 2011.
- Diener, E., S. J. Heintzelman, K. Kushlev, L. Tay, D. Wirtz, L. D. Lutes, and S. Oishi, "Findings All Psychologists Should Know From the New Science on Subjective Well-Being," *Canadian Psychology/Psychologie canadienne*, 58, 87–104, 2017.
- Dolan, P., T. Peasgood, and M. White, "Do We Really Know What Makes Us Happy? A Review of the Economic Literature on the Factors Associated with Subjective Well-Being," *Journal of Economic Psychology*, 29, 94–122, 2008.
- Di Tella, R., J. Haisken-De New, and R. MacCulloch, "Happiness Adaptation to Income and to Status in an Individual Panel," *Journal of Economic Behavior & Organization*, 76, 834–52, 2010.
- Di Tella, R., R. J. MacCulloch, and A. J. Oswald, "Preferences over Inflation and Unemployment: Evidence from Surveys of Happiness," *American Economic Review*, 91, 335–41, 2001.
- Dunn, E., L. Aknin, and M. Norton, "Spending Money on Others Promotes Happiness," Science, 319, 1687–88, 2008.
- Dustmann, C. and F. Fasani, "The Effect of Local Area Crime on Mental Health," *Economic Journal*, 126, 978–1017, 2016.
- Easterlin, R., "Does Economic Growth Improve the Human Lot?," in P. A. David and W. B. Melvin, eds, *Nations and Households in Economic Growth*, 89–125, Stanford University Press, Palo Alto, 1974.
 - —, "Paradox Lost?," *Review of Behavioral Economics*, 4, 311–39, 2017.
- Epel, E., "Telomeres in a Life-Span Perspective: A New "Psychobiomarker"?," Current Directions in Psychological Science, 18, 6–10, 2009.

Ferreira, S. and M. Moro, "On the Use of Subjective Well-Being Data for Environmental Valuation," Environmental and Resource Economics, 46, 249–73, 2010.

Ferrer-i-Carbonell, A., "Income and Well-Being: An Empirical Analysis of the Comparison Income Effect," *Journal of Public Economics*, 89, 997–1019, 2005.

- Ferrer-i-Carbonell, A. and J. Gowdy, "Environmental Degradation and Happiness," *Ecological Economics*, 60, 509–16, 2007.
- Flavin, P. and M. Keane, "Life Satisfaction and Political Participation: Evidence from the United States," *Journal of Happiness Studies*, 13, 63–78, 2012.
- Flavin, P., A. Pacek, and B. Radcliff, "Labor Unions and Life Satisfaction: Evidence From New Data," *Social Indicators Research*, 98, 435-49, 2010.

——, "Assessing the Impact of the Size and Scope of Government on Human Well-Being," *Social Forces*, 92, 1241–58, 2014.

Flèche, S., "Distaste for Centralization: Evidence from a Quasi-Natural Experiment in Switzerland," CEP Discussion Paper No 1383, 2015.

——, "Teacher Quality, Test Scores and Non-Cognitive Skills: Evidence from Primary School Teachers in the UK," CEP Discussion Paper No 1472, 2017.

Fließbach, K., B. Weber, P. Trautner, T. Dohmen, U. Sunde, C. Elger, and A. Falk, "Social comparison affects reward-related brain activity in the human ventral striatum," *Science*, 318, 1305–08, 2007.

- Frank, R. H. and R. M. Hutchens, "Wages, Seniority, and the Demand for Rising Consumption Profiles," *Journal of Economic Behavior and Organization*, 21, 251–76, 1993.
- Freeman, R. B., "Job Satisfaction as an Economic Variable," American Economic Review, 68, 135–41, 1978.
- Frey, B. S. and M. Benz, "Being Independent Is a Great Thing: Subjective Evaluations of Self-Employment and Hierarchy," *Economica*, 75, 362–83, 2008.
- Frey, B. S., S. Luechinger, and A. Stutzer, "The Life Satisfaction Approach to the Value of Public Goods: The Case of Terrorism," *Public Choice*, 138, 317–45, 2009.
- Frey, B. S. and A. Stutzer, "Happiness, Economy and Institutions," *Economic Journal*, 110, 918–38, 2000.
 ——, "What Can Economists Learn from Happiness Research?," *Journal of Economic Literature*, 40, 402–35, 2002.
- Gardner, J. and A. J. Oswald, "Money and Mental Wellbeing: A Longitudinal Study of Medium-Sized Lottery Wins," *Journal of Health Economics*, 26, 49–60, 2007.
- Green, C., J. Heywood, P. Kler, and G. Leeves, "Paradox Lost: Disappearing Female Job Satisfaction," British Journal of Industrial Relations, forthcoming, 2018.
- Grün, C., W. Hauser, and T. Rhein, "Is Any Job Better than No Job? Life Satisfaction and Re-employment," *Journal of Labor Research*, 31, 285–306, 2010.
- Grund, C. and D. Sliwka, "Reference-Dependent Preferences and the Impact of Wage Increases on Job Satisfaction: Theory and Evidence," *Journal of Institutional and Theoretical Economics*, 163, 313–35, 2007.
- Hamermesh, D. S., "Economic Aspects of Job Satisfaction," in O. C. Ashenfelter and W. E. Oates, eds, *Essays in Labor Market Analysis*, 53–72, John Wiley, New York, 1977.
- Hamermesh, D., D. Kawaguchi, and J. Lee, "Does Labor Legislation Benefit Workers? Well-Being after an Hours Reduction," *Journal of the Japanese and International Economies*, 44, 1–12, 2017.
- Helliwell, J. and S. Wang, "The State of World Happiness," in J. Helliwell, R. Layard, and J. Sachs, eds, World Happiness Report, 10–57, Columbia Earth Institute, New York, 2012.
- Helliwell, J. F. and H. Huang, "How's Your Government? International Evidence Linking Good Government and Well-Being," *British Journal of Political Science*, 38, 595–619, 2008.
- Helliwell, J. F., H. Huang, and S. Wang, "New Evidence on Trust and Well-Being," in Uslaner, E., ed., The Oxford Handbook of Social and Political Trust, Oxford University Press, Oxford, 2018.
- Jahoda, M., Employment and Unemployment: A Social-Psychological Analysis, Cambridge: Cambridge University Press, 1982.
- Kahneman, D. and A. Deaton, "High Income Improves Evaluation of Life but not Emotional Well-Being," *Proceedings of the National Academy of Science*, 107, 16489–493, 2010.
- Knabe, A. and S. Rätzel, "Scarring or Scaring? The Psychological Impact of Past Unemployment and Future Unemployment Risk," *Economica*, 78, 283–93, 2011.
- Krekel, C., J. Kolbe, and H. Wuestemann, "The Greener, The Happier? The Effect of Urban Land Use on Residential Well-Being," *Ecological Economics*, 121, 117–27, 2016.
- Layard, R., A. E. Clark, F. Cornaglia, N. Powdthavee, and J. Vernoit, "What Predicts a Successful Life? A Life-Course Model of Wellbeing," *Economic Journal*, 124, F720–F738, 2014.
- Layard, R., A. E. Clark, and C. Senik, "The Causes of Happiness and Misery," in J. Helliwell, R. Layard, and J. Sachs, eds, *World Happiness Report 2012*, 58–89, Columbia Earth Institute, New York, 2012.
- Layard, R., G. Mayraz, and S. Nickell, "Does Relative Income Matter? Are the Critics Right?," in E. Diener, J. Helliwell, and D. Kahneman, eds, *International Differences in Well-Being*, 436–68, Oxford University Press, Oxford, 2010.
- Lepinteur, A., "The Shorter Workweek and Worker Wellbeing: Evidence from Portugal and France," PSE Working Paper No. 2016-21, 2016.
- Liberini, F., M. Redoano, and E. Proto, "Happy Voters," Journal of Public Economics, 146, 41–57, 2017.
- Loewenstein, G. and N. Sicherman, "Do Workers Prefer Increasing Wage Profiles?," Journal of Labor Economics, 9, 67–84, 1991.
- Lora, E., Beyond Facts. Understanding Quality of Life, IDB Press, 2008.
- Luechinger, S., "Valuing Air Quality Using the Life Satisfaction Approach," *Economic Journal*, 119, 482–515, 2009.
- Luechinger, S., S. Meier, and A. Stutzer, "Bureaucratic Rents and Life Satisfaction," Journal of Law, Economics and Organization, 24, 476–88, 2008.
- Luttmer, E., "Neighbors as Negatives: Relative Earnings and Well-Being," *Quarterly Journal of Economics*, 120, 963–1002, 2005.
- Metcalfe, R., P. Dolan, and N. Powdthavee, "Destruction and Distress: Using a Quasi-Experiment to Show the Effects of the September 11 Attacks on Subjective Well-Being in the UK," *Economic Journal*, 121, F81–F103, 2011.

- Moore, S. and J. Shepherd, "The Elements and Prevalence of Fear," The British Journal of Criminology, 47, 154–62, 2007.
- Nikolaev, B., "Does Other People's Education Make Us Less Happy?," Economics of Education Review, 52, 176–91, 2016.
- Nolen-Hoeksema, S. and C. L. Rusting, "Gender Differences in Well-Being," in D. Kahneman, E. Diener, and N. Schwartz, eds, *Well-being: The foundations of hedonic psychology*, 213–29, Russell Sage Foundation, New York, 1999.
- Oishi, S., U. Schimmack, U., and E. Diener, "Progressive Taxation and the Subjective Well-Being of Nations," *Psychological Science*, 23, 86–92, 2004.
- Oreopoulos, P. and K. Salvanes, "Priceless: the Nonpecuniary Benefits of Schooling," Journal of Economic Perspectives, 25, 159–84, 2011.
- Oswald, A. J., "Happiness and Economic Performance," Economic Journal, 107, 1815-31, 1997.
- Oswald, A. J., E. Proto, and D. Sgroi, "Happiness and Productivity," *Journal of Labor Economics*, 33, 789–822, 2015.

Oswald, A. and S. Wu, "Well-Being across America," Review of Economics and Statistics, 93, 1118–134, 2011.

- Pacek, A. and B. Radcliff, "Welfare Plicy and Subjective Well-Being Across Nations: An Individual-Level Assessment," Social Indicators Research, 89, 179–91, 2008.
- Pischke, S., "Money and Happiness: Evidence from the Industry Wage Structure," LSE, mimeo, 2010.
- Platt, S., R. Micciolo, R., and M. Tansella, "Suicide and Unemployment in Italy: Description, Analysis and Interpretation of Recent Trends," *Social Science & Medicine*, 34, 1191–201, 1992.
- Pluess, M., Genetics of Psychological Well-Being: The Role of Heritability and Genetics in Positive Psychology, Oxford University Press, Oxford, 2015.
- Powdthavee, N., "Unhappiness and Crime: Evidence from South Africa," Economica, 72, 531-47, 2005.
- Rehdanz, K. and D. Maddison, "Climate and Happiness," Ecological Economics, 52, 111-25, 2005.
- Rehdanz, K., H. Welsch, D. Naritaa, and T. Okubo, "Well-being Effects of a Major Natural Disaster: The Case of Fukushima," *Journal of Economic Behavior and Organization*, 116, 500–17, 2015.
- Solnick, S. and D. Hemenway, "Are Positional Concerns Stronger in Some Domains than in Others?," American Economic Review, 95, 147–51, 2005.
- Sousa-Poza, A. and A. Sousa-Poza, "Gender Differences in Job Satisfaction in Great Britain, 1991– 2000: Permanent or Transitory?," *Applied Economics Letters*, 10, 691å4, 2003.
- Stevenson, B. and J. Wolfers, "Happiness Inequality in the United States," *Journal of Legal Studies*, 37, S33–S79, 2008.
- ——, "The Paradox of Declining Female Happiness," *American Economic Journal: Economic Policy*, 1, 190–225, 2009.
- Stutzer, A. and B. S. Frey, "Does Marriage Make People Happy, Or Do Happy People Get Married?," Journal of Socio-Economics, 35, 326–47, 2006.
- Thoits, P. and L. Hewitt, "Volunteer Work and Well-Being," *Journal of Health and Social Behavior*, 42, 115–31, 2001.
- Urry, H., J. Nitschke, I. Dolski, D. Jackson, K. Dalton, C. Mueller, M. Rosenkranz, C. Ryff, B. Singer, and R. Davidson, "Making a Life Worth Living: Neural Correlates of Well-Being", *Psychological Science*, 15, 367–72, 2004.
- Van de Stadt, H., A. Kapteyn, and S. Van de Geer, "The Relativity of Utility: Evidence from Panel Data," *Review of Economics and Statistics*, 67, 179–87, 1985.
- Van Praag, B. M. S., "The Welfare Function of Income in Belgium: An Empirical Investigation," *European Economic Review*, 2, 337–69, 1971.
- Van Praag, B. M. S. and B. E. Baarsma, "Using Happiness Surveys to Value Intangibles: the Case of Airport Noise," *Economic Journal*, 115, 224-46, 2005.
- Vendrik, M., "Adaptation, Anticipation and Social Interaction in Happiness: An Integrated Error-Correction Approach," *Journal of Public Economics*, 105, 131-49, 2013.
- Ward, G., "Is Happiness a Predictor of Election Results?," CEP Discussion Paper No. 1343, 2017.
- Weiss, A., J. King, M. Inoue-Murayama, T. Matsuzawa, and A. Oswald, "Evidence for a Midlife Crisis in Great Apes Consistent with the U-shape in Human Well-Being," *Proceedings of the National Academy of Sciences of the USA*, 109, 19949–952, 2012.
- Winkelmann, L. and R. Winkelmann, "Why are the Unemployed so Unhappy? Evidence from Panel Data," *Economica*, 65, 1–15, 1998.
- Zizzo, D. J. and A. J. Oswald, "Are People Willing to Pay to Reduce Others' Incomes?," *Annales d'Economie et de Statistique*, 63–64, 39–65, 2001.